Author's response to reviews

Title: Becoming a general practitioner? - Which factors have main impact on career choice of medical students

Authors:

Kathrin Kiolbassa (Kathrin.Kiolbassa@gmx.de)
Antje Miksch (antje.miksch@med.uni-heidelberg.de)
Katja Hermann (katja.hermann@med.uni-heidelberg.de)
Andreas Loh (Andreas.Loh@uniklinik-freiburg.de)
Joachim Szecsenyi (joachim.szecsenyi@med.uni-heidelberg.de)
Stefanie Joos (stefanie.joos@med.uni-heidelberg.de)
Katja Goetz (katja.goetz@med.uni-heidelberg.de)

Version: 2 Date: 25 October 2010

Author's response to reviews:

BioMed Central Editorial Team,

regarding your mail from 21th October we will submit the two additional files of our manuscript: Becoming a general practitioner? - Which factors have main impact on career choice of medical students to BMC Family Practice.

Please find two additional files, a translated copy of the questionnaire and the solution to the factor analysis attached.

King regards

Katja Goetz